Author’s response to reviews

Title: An efficacy and mechanism evaluation study of Levosimendan for the Prevention of Acute oRgan Dysfunction in Sepsis (LeoPARDS): protocol for a randomised controlled trial

Authors:

Robert M. L’E Orme (robert.orne@glos.nhs.uk)
Gavin D. Perkins (G.D.Perkins@warwick.ac.uk)
Daniel F. McAuley (d.f.mcauley@qub.ac.uk)
Kathleen D. Liu (Kathleen.liu@ucsf.edu)
Alexina J. Mason (andrea.mason05@imperial.ac.uk)
Andrea Morelli (andrea.morelli@uniroma1.it)
Mervyn Singer (m.singer@ucl.ac.uk)
Deborah Ashby (deborah.ashby@imperial.ac.uk)
Anthony C. Gordon (anthony.gordon@imperial.ac.uk)

Version: 4
Date: 14 May 2014

Author’s response to reviews:

Dear Editorial team
Thank you for your email. We have addressed the two minor formatting requests.
1. In the abstract we have added the date of trial registration (19 September 2013).
2. We have included a discussion. There is now a short discussion in the abstract and in the main text.
These changes are shown in red on the new uploaded version.
Thank you for considering this revised version for publication.

Yours sincerely
Anthony Gordon